Author’s response to reviews

Title: Study Protocol of a Randomized Controlled Trial to Test the Effect of a Smartphone Application on Oral-health behavior and Oral Hygiene in Adolescents with Fixed Orthodontic Appliances.

Authors:

Janneke Scheerman (j.f.m.scheerman@acta.nl)

Berno Van Meijel (Berno.vanMeijel@inholland.nl)

Pepijn van Empelen (Pepijn.vanEmpelen@tno.nl)

Gem Kramer (G.j.c.Kramer@acta.nl)

Gijsbert Verrips (Erik.Verrips@tno.nl)

Amir Pakpour (Pakpour_Amir@yahoo.com)

Matheus van den Braak (t.vd.braak@hotmail.com)

Cor van Loveren (c.van.Loveren@acta.nl)

Version: 3 Date: 11 Jan 2018

Author’s response to reviews:

J. F.M. Scheerman MSc.

PhD-student and Lecturer

Academic Centre of Dentistry Amsterdam,

Department of Preventive Dentistry,

Gustav Mahlerlaan 3004, 1081 LA,

Amsterdam

The Netherlands

Tel: 00316 30899701

Email: j.f.m.scheerman@acta.nl;

janneke.scheerman@inholland.nl.
Amsterdam, January 2018, 11th

Concerning: rebuttal letter of the revised manuscript ID OHEA-D-17-00008R2

Dear Editor,

Thank you for considering our manuscript for publication in BMC Oral Health. Please find enclosed our revised manuscript entitled: “Study Protocol of a Randomized Controlled Trial to Test the Effect of a Smartphone Application on Oral-health Behavior and Oral Hygiene in Adolescents with Fixed Orthodontic Appliances.” (Manuscript ID OHEA-D-17-00008R2).

We would like to thank the editor, Robert Jones, Fernando Hugo for reviewing the manuscript. We have included the information about the funding in the manuscript.

Sincerely yours, also on behalf of the co-authors,

Janneke F.M. Scheerman